Title: Evaluation of a strict protocol approach in managing women with severe disease due to hypertension in pregnancy: A before and after study.

Authors:

Hennie Lombaard (hennie.lombaard@up.ac.za)
Robert C Pattinson (rcpattin@kalafong.up.ac.za)
Febe Backer (febe.backer@up.ac.za)
Peter Macdonald (petermac@medic.ac.za)

Version: 6 Date: 27 September 2005

Author’s response to reviews:

editing changes